Author's response to reviews

Title: Estrogen receptor alpha (ERalpha) mRNA copy numbers in immunohistochemically positive-, and negative breast cancer tissues

Authors:

Indira Poola (ipoola@howard.edu)
Qingqi Yue (yueqingqi@yahoo.com)

Version: 6 Date: 14 March 2007

Author's response to reviews:

March 14, 2007
To

The Editor
BMC Cancer

REF:
Manuscript # 1961808621131316

Title: Estrogen receptor alpha (ERalpha) mRNA copy numbers in immunohistochemically positive-, and negative breast cancer tissues

Authors: I. Poola and Q. Yue

Dear Editor,
Thank you for the email to let me know about the core acceptance of our above manuscript.

As requested all the editorial corrections have been made and the revised version is uploaded.

Sincerely,

Indira Poola, Ph.D
Professor of Biochemistry and Molecular Biology